Author’s response to reviews

Title: Knockdown of estrogen receptor β increases proliferation and affects the transcriptome of endometrial adenocarcinoma cells

Authors:

Oliver Treeck (otreeck@caritasstjosef.de)
Elisabeth Diepolder (elisabeth.diepolder@web.de)
Maciej Skrzypczak (skrzypm@yahoo.co.uk)
Susanne Schüler-Toprak (sschueler@caritasstjosef.de)
Olaf Ortmann (oortmann@caritasstjosef.de)

Version: 1 Date: 27 Nov 2018

Author’s response to reviews:

Dear Sir or Madam, we made all revisions you suggested particularly with regard to the mentioned text overlap in the methods section.

With kind regards Oliver Treeck